Author's response to reviews

Title: Prognostic value of physicians assessment of compliance regarding all-cause mortality in patients with type 2 diabetes: primary care follow-up study

Authors:

Dietrich Rothenbacher (d.rothenbacher@dkfz.de)
Gernot Ruter (rueter@telemed.de)
Hermann Brenner (h.brenner@dkfz.de)

Version: 5 Date: 6 June 2006

Author's response to reviews: see over
MS: 1438626914874219 - #Prognostic value of physicians assessment of compliance regarding all-cause mortality in patients with type 2 diabetes: primary care follow-up study'

Dear Sir,

thank you for your e-mail dated June 2, 2006 in which you asked us for further formatting changes of the above named manuscript to be published in "BMC Family Practice ".

We did the suggested formatting changes. However, we kept the order of manuscript sections as indicated in the "Instructions for Authors" of “BMC Family Practice”.

Yours sincerely,

Dietrich Rothenbacher, MD, MPH